Author's response to reviews

Title: Maximising the impact of qualitative research in feasibility studies for randomised controlled trials: guidance for researchers

Authors:

Alicia O’Cathain (a.ocathain@sheffield.ac.uk)
Pat Hoddinott (p.m.hoddinott@stir.ac.uk)
Simon Lewin (simon.lewin@nokc.no)
Kate J Thomas (k.thomas@sheffield.ac.uk)
Bridget Young (Bridget.Young@liverpool.ac.uk)
Joy Adamson (joy.adamson@york.ac.uk)
Yvonne JFM Jansen (yvonne.jansen@tno.nl)
Nicola Mills (nicola.mills@bristol.ac.uk)
Graham Moore (MooreG@cardiff.ac.uk)
Jenny L Donovan (jenny.donovan@bristol.ac.uk)

Version: 4 Date: 11 August 2015

Author's response to reviews:

Hi
We have added three case studies to the paper as requested. Each addresses a number of items from the whole guidance rather than being domain specific.
Best wishes
Alicia